Author’s response to reviews

Title: Leveraging the EHR4CR Platform to Support Patient Inclusion in Academic Studies: Challenges and Lessons Learned

Authors:
yannick girardeau (yannick.girardeau@aphp.fr)
Justin Doods (justin.doods@ukmuenster.de)
Eric Zapletal (eric.zapletal@aphp.fr)
Gilles Chatellier (gilles.chatellier@aphp.fr)
Christel Daniel (christel.daniel@aphp.fr)
Anita Burgun (anita.burgun@aphp.fr)
Martin Dugas (martin.dugas@ukmuenster.de)
Bastien Rance (bastien.rance@aphp.fr)

Version: 2 Date: 27 Dec 2016

Author’s response to reviews:

Dear Editor,

Please find below our point by point response to the comments.

Best regards,

Dr. Yannick Girardeau

1. Please edit the title of the manuscript so that it is in sentence case with only the first word and the acronyms capitalised.

Done, line 1 to 3

2. Please move all tables to the end of the main text.

Done, line 448 to 462
3. Please incorporate the materials section into the methods so that the manuscript has the following sections: background, methods, results, discussion and conclusions.

Done

4. Please copy edit the entire manuscript and ensure that all spelling and grammar errors are corrected. E.g. page 6, line 130, terminologies to terminologies.

Done, page 6, line 130, terminologies to terminologies and decribed to described